Author's response to reviews

Title: Relationship between visual field loss and contrast threshold elevation in glaucoma

Authors:

C M Tochel (jdm1u@udcf.gla.ac.uk)
J S Morton (jdm1u@udcf.gla.ac.uk)
J L Jay (jdm1u@udcf.gla.ac.uk)
J D Morrison (jdm1u@udcf.gla.ac.uk)

Version: 4 Date: 29 August 2005

Author's response to reviews:

Formatting changes have been made as specified. The abbreviations given referred specifically to Table 1 and not to the MS as a whole and have been retained under that Table. If the Tables are not exactly as you would wish, they can be provided at TEX files.